Author's response to reviews

Title: Plasma 24S-hydroxycholesterol levels in elderly subjects with late onset Alzheimer's disease or vascular dementia: a case-control study

Authors:

GIOVANNI ZULIANI (gzuliani@hotmail.com)
MICHELA PERRONE DONNORSO (michela.perrone.donnorso@uniroma2.it)
CRISTINA BOSI (c.bosi@ospfe.it)
ANGELINA PASSARO (a.passaro@ospfe.it)
EDOARDO DALLA NORA (e.dallanora@ospfe.it)
AMEDEO ZURLO (a.zurlo@ospfe.it)
FRANCESCO BONETTI (francesco.bonetti@virgilio.it)
ALESSIA MOZZI (alessia.francesca.mozzi@uniroma2.it)
CLAUDIO CORTESE (claudio.cortese@uniroma2.it)

Version: 4 Date: 5 September 2011

Author's response to reviews:

Dear Dr. A.K. Nair,

please find enclosed the revised version of the manuscript. We have modified the second sentence of the Discussion as requested, and now it is more clear to the reader (C: controls).

Waiting for your kin answer.

Best regards.

Giovanni Zuliani MD PhD